Author’s response to reviews

Title: Pregnancy denial and early infant development: a case-control prospective study.

Authors:

Julie Auer (jauer@chu-reims.fr)
Coralie Barbe (cbarbe@chu-reims.fr)
Anne-Laure Sutter (alsutter@ch-perren.fr)
Dominique Dallay (dominique.dallay@chu-bordeaux.fr)
Laurianne Vulliez (l.vulliez@chu-besancon.fr)
Didier Riethmuller (didier.riethmuller@univ-fcomte.fr)
Violaine Gubler (violaine.gubler@gmail.com)
Valérie Verlome (valérie.verlomme@ch-toulon.fr)
Stéphanie Saad-Saint-Gilles (stephanie.saad-saint-gilles@cpn-laxou.com)
Alain Miton (A.MITON@chru-nancy.fr)
Emmanuelle Tessier (tessier.e@chu-toulouse.fr)
Olivier Parant (parant.o@chu-toulouse.fr)
Julie Le Foll (julie.le-foll@bch.aphp.fr)
Agnès Bourgeois-Moine (agnes.bourgeois-moine@aphp.fr)
Sylvie Viaux (sylvie.viaux@psl.aphp.fr)
Marc Dommergues (marc.dommergues@psl.aphp.fr)
Gisèle Apter (gisdanap@aol.com)
Joëlle Belaisch-Allart (j.belaischallart@ch4v.fr)
Anne Danion (Anne.Danion@chru-strasbourg.fr)
Israël Nisand (israel.nisand@chru-strasbourg.fr)
Dear BMC Editor,

Please find here the response to your last comments

1. Ethical and Funding Approval Documentation

We just sent an email with the relevant documents concerning Ethical and Funding Approval to BMCSeriesEditorial@biomedcentral.com and benjamin.ragen@biomedcentral.com.

2. Funding

Our study protocol has received a governmental funding from the French Ministry for Solidarity and Health after having been evaluated by the PHRCN 2011 jury’s panel (peer-reviews).

3. Study status

Our study is currently ongoing, all the inclusions are done but we are still doing the follow up (on 20 months) of the mothers and children by collecting and analysing data.

4. Related articles

We certify that the submission is original work and is not under review for publication with any other journals. Moreover no primary results have been published or submitted to a journal.

Concerning the English language our manuscript has been reviewed and certified by AJE (American Journal Experts).

Yours sincerely,

Pr AC Rolland on behalf of the authors.